Author’s response to reviews

Title: Mechanisms and pathways to impact in public health research: A preliminary analysis of research funded by the National Institute for Health Research (NIHR)

Authors:
Harriet Boulding (harriet.boulding@kcl.ac.uk)
Adam Kamenetzky (adam.kamenetzky@kcl.ac.uk)
Ioana Ghiga (ighiga@rand.org)
Rebecca Ioppolo (rioppolo@rand.org)
Facundo Herrera (fherrera@rand.org)
Sarah Parks (sparks@rand.org)
Catriona Manville (manville@rand.org)
Susan Guthrie (sguthrie@rand.org)
Saba Hinrichs-Krapels (sabzie@gmail.com; saba.hinrichs@kcl.ac.uk)

Version: 2 Date: 16 Jan 2020

Author’s response to reviews:

Dear editorial team,

We are pleased to submit our revised manuscript and have made the following minor changes in response to the editorial and reviewer comments received.

Please respond to the reviewer comment.

We have responded to the last remaining reviewer comment (to add the interview topic guide). This is now included as supplementary material and a description of how this was inspired in the methods section is on pg 5 last paragraph.
Please include a statement in the Authors’ contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

All authors have read the manuscript and a statement to this effect is now in the final version.

Please include a copy (in English) of your questionnaire in the supplementary files or a reference to it if it is already published.

As above, this is now included as a supplementary material.

If you wish to acknowledge someone by name in the Acknowledgements, please ensure you have obtained permission from them to so do.

Staff at NIHR who helped with the analysis have been mentioned by name and were happy to be acknowledged.

We thank the editorial team for their support in the submission of our manuscript.

With kind regards

Saba Hinrichs-Krapels, on behalf of the authors